Author’s response to reviews

Title: A case report of Dengue Haemorrhagic fever complicated with psoas haematoma requiring blood transfusion

Authors:

Anne Thushara Matthias (thushara.matthias@gmail.com)
Sanduni Apsara (Sanduniap88@gmail.com)
Apsara Epa (apsaraepa@yahoo.com)

Version: 1 Date: 22 Sep 2018

Author’s response to reviews:

Response to reviewers

1. Care check list is attached

2. As ethics approval was not need as this was a case report, no formal ethics review application was submitted. It was verbally clarified ethics approval was not needed.

3. Verbal and written consent was taken form the patient for publication.

4. All authors have read and approved the final manuscript.